Author’s response to reviews

Title: A prospective study on elective surgical inpatient satisfaction with perioperative anaesthesia service at Ayder comprehensive specialized hospital, Mekelle, Ethiopia

Authors:
Kore Benwu (lulakm2020@gmail.com)
Hagos Gebermedhin (hagosg484@gmail.com)

Version: 2 Date: 09 Nov 2018

Author’s response to reviews:

Most of the comments were acceptable and major revision is done on the different section of the document

The references issue in the introduction and discussion section is now corrected

The figure description document which was found at the end of the manuscript is moved to result part of the main document

Most of the spellings are changed to right words

The editor had comments on the analysis part and it is done according to his suggestion

And most of the suggestion from the editor was acceptable and we incorporate to the revision.